Author's response to reviews

Title: Four minutes for a patient, twenty seconds for a relative - an observational study at a university hospital

Authors:

Gerhild Becker (gerhild.becker@uniklinik-freiburg.de)
Dorothee E Kempf (dorothee.kempf@uniklinik-freiburg.de)
Carola J Xander (carola.xander@uniklinik-freiburg.de)
Felix Momm (felix.momm@uniklinik-freiburg.de)
Manfred Olschewski (mol@imbi.uni-freiburg.de)
Hubert E Blum (hubert.blum@uniklinik-freiburg.de)

Version: 8 Date: 2 April 2010

Author's response to reviews:

Three minor revisions have been done:
1. p 6: a . has been added after the term Tabel 3
2. p 14: the second part of the title has now also been written in bold letters
3. a break has been pasted between Table 1 and table 2